Author's response to reviews

Title: A population-based survey on tanning bed use in Germany

Authors:

Franziska U Börner (f.boerner@fz-juelich.de)
Holger Schütz (h.schuetz@fz-juelich.de)
Peter Wiedemann (p.wiedemann@fz-juelich.de)

Version: 4 Date: 6 July 2009

Author’s response to reviews:

July 6th, 2009

Dear Editor,

we apologize for the missing information on the ethical approval and have added the ethical approval information in our document. Thank you for considering our paper for publication.

Sincerely,

Franziska Börner